Author's response to reviews

Title: Incidence, Mortality Patterns and 5-year Relative Survival Ratio of Prostate Cancer among Residents in Singapore from 1968 to 2002

Authors:

Sin Eng Chia (cofcse@nus.edu.sg)
Chuen Seng Tan (cmetcs@nus.edu.sg)
Gek Hsiang Lim (cmelgh@nus.edu.sg)
Xueling Sim (cmesx@nus.edu.sg)
Yudi Pawitan (Yudi.Pawitan@ki.se)
Marie Reilly (Marie.Reilly@ki.se)
Safiyya Mohamed Ali (safiyya@nus.edu.sg)
Weber Lau (weber.lau.k.o@sgh.com.sg)
Kee Seng Chia (cofcks@nus.edu.sg)

Version: 6 Date: 9 September 2008

Author's response to reviews:

Dear Editor,

We have included what you have requested in the revised manuscript. Below is what we have included:

Acknowledgements:

This study was supported in part by a grant from the National Medical Research Council, NMRC/EDG/0008/2007. We thank the Singapore Cancer Registry for permission to use their data. This study has been approved by the National University of Singapore Institutional Review Board (Approval number 06-062).